Author's response to reviews

Title: A school-based resilience intervention to decrease tobacco, alcohol and marijuana use in high school students

Authors:

Rebecca K Hodder (rebecca.hodder@hnehealth.nsw.gov.au)
Justine Daly (justine.daly@hnehealth.nsw.gov.au)
Megan Freund (megan.freund@hnehealth.nsw.gov.au)
Jenny Bowman (jenny.bowman@newcastle.edu.au)
Trevor Hazell (trevor.hazell@hnehealth.nsw.gov.au)
John Wiggers (john.wiggers@hnehealth.nsw.gov.au)

Version: 4 Date: 31 May 2011

Author's response to reviews:

Dear Dr Silvia Martins

Please find attached a revised version of the manuscript. The following changes have been made:

- The title in the submission system has been amended to reflect that in the manuscript
- All highlighting/tracking has been removed from the document

Regards
Rebecca